\begin{table}%t2
%\begin{center}
\par
\caption{References for the uncorrected photometry used in Figs.\ \ref{SED1} to~\ref{SED6}.}
\label{Photometry}
\par
\begin{tabular}{lccc}
\hline
\hline\noalign{\smallskip}
{ Star } &  { {\it UBV}} &   { ({\it RI})$_{\rm{c}}$} & {\it JHKL} \\\noalign{\smallskip}
\hline
\par
     T Cha &          1 &         1  &          1/2 \\
     Sz 68 &          3 &         4  &          2/5 \\
    RU Lup &          6 &         6  &          2/6 \\
    EX Lup &        7/8 &       8/9  &          2/8 \\
    Sz 124 &         10 &         8  &          2/8 \\
     ROX 2 &         11 &        11  &          2 \\
   DoAr 24 &         11 &        11  &          2/12 \\
  DoAr 24E &         11 &      13$^{kc}$  &         2/12 \\
     SR 12 &         12 &      13$^{kc}$ &         2/12 \\
    ROX 31 &         11 &        11  &          2/14 \\
  Wa Oph/4 &         15 &      16/17  &          2 \\
  Wa Oph/6 &         15 &          --  &          2 \\
    IX Oph &          4 &         4  &          2/18 \\
 V1725 Oph &         19 &          --  &          2 \\
CoKu Ser G1 &        20 &         20  &          2 \\
CoKu Ser G7 &     20/21 &         20  &          2 \\
  V536 Aql &         20 &         20  &          2/5 \\
   MacC H3 &          3 &      16/20  &          2 \\
\par
\hline
\end{tabular}
\par
%\end{center}
\par
%\caption{References for the photometry used in the graphs.
%\newline
%\footnote{
Notes to Table:
kc~=~photometry in the Cousins system,
after being transformed from the Kunkel and Rydgren system using the relationships in \cite{Bessell79}.
\newline
{
\textit{References:} (1)~\cite{Alcala93}, (2)~2MASS, (3)~Herbig-Bell Catalogue (HBC), (4)~\cite{Covino92},
(5)~\cite{Geoffray01},
(6)~\cite{Gahm93}, (7)~\cite{Mundt80}, (8)~\cite{Hughes94}, (9)~\cite{Herbig92}, (10)~\cite{Schwartz78},
(11)~\cite{Bouvier92}, (12)~\cite{Rydgren76}, (13)~\cite{Rydgren80}, (14)~\cite{Ageorges97}, (15)~\cite{Walter86},
(16)~USNO-A2.0, (17)~DENIS
(18)~\cite{Glass74}, (19)~\cite{Reipurth86}, (20)~Observations for this paper taken 
using the Isaac Newton Telescope on La Palma, 
(21)~\cite{Cohen79}.                                   
}
\end{table}